Author's response to reviews

Title: Which dimensions of psychosocial working conditions are monitored in Europe?

Authors:

Maren Formazin (formazin.maren@baua.bund.de)
Hermann Burr (burr.hermann@baua.bund.de)
Cecilie Aagestad (cecilie.aagestad@stami.no)
Tore Tynes (tore.tynes@stami.no)
Sannie Vester Torsen (svt@arbejdsmiljoforskning.dk)
Merja Perkio-Makela (Merja.Perkio-Makela@ttl.fi)
Clara Isabel Diaz Aramburu (cdiazara@mtin.es)
Javier Pinilla Garcia (jpinilla@mtin.es)
Luz Galiana Blanco (luzgal@mtin.es)
Greet Vermeylen (Greet.Vermeylen@eurofound.europa.eu)
Agnes Parent-Thirion (Agnes.Parent-Thirion@eurofound.europa.eu)
Wendela Hooftman (wendela.hooftman@tno.nl)
Irene Houtman (irene.houtman@tno.nl)

Version: 2
Date: 27 June 2014

Author’s response to reviews:

Dear editor,

thank you very much for your quick reply to our submission and for pointing out the changes necessary for proceeding with peer review.

We have changed the following:

1. We have added a section “Ethical approval” at the end of our manuscript (p.17, lines 412 to 414) in which we have pointed out that ethical approval was not necessary because we analysed the content of questionnaires (and not participant data).

2. We have added a section “Acknowledgements” at the end of our manuscript (p.18, lines 422 to 426) in which we have added our thanks to Beate Beermann, Anke Siefer and Hans Martin Hasselhorn for supplying information on the German BIBB-BAuA questionnaire and for valuable feedback on the draft of the manuscript. In this section, we additionally point out that the work of the group has not been funded by outside partners, but was a part of all the participating institutes’ ordinary activities.

We thank you very much for your kind review of our manuscript.

Best regards,

On behalf of all co-authors
Maren Formazin